CENWP-OP-B                                                                                                                                          29 July 2003

MEMORANDUM FOR CENWP-OP, ATTN: Gary Johnson

SUBJECT: Bonneville Lock and Dam, Fishway and Fish Activities for week 30 of 2003, which covers the period 20 through 26 July 2003.

1.  OPERATION SUMMARY.

1.1 Daily river flow averaged 144.0 to 186.1 kcfs.  Daily project tailwater elevation averaged 11.4’ to 14.5’ msl.  Daily project forebay elevation averaged 75.5’ to 75.7’ msl.  River temperature averaged 67 to 69° F.  Daily turbidity measured 5.0’ to 7.0’ on the secchi disk.

1.2 Spill Bays 1 and 18, which supply attraction flow for Cascade Island and Bradford Island fishways respectively, are included in the spill pattern.  Spill averaged 84.8 to 100.2 kcfs.  

1.3 Unit Operation

1.3.1 Unit 1 remains out of service for turbine rehab.

1.3.2 Unit 7 was out of service from 0930 until 1106 on 22 July for asbestos sampling.

1.3.3 Unit 9 had STSs installed at 1630 on 23 July.

1.3.4 Unit 14 was out of service from 1130 until 1338 on 22 July for asbestos sampling.

1.3.5 Unit 15 was out of service from 0400 on 21 July until 1100 on 22 July due to a bearing oil leak.

1.3.6 Unit 17 was out of service from 0500 until 1736 on 23 July for annual maintenance.  Unit 17 was also out of service from 1030 until 2200 on 25 July due to difficulty starting from GDACS.

1.4 Second Powerhouse fish units provide attraction flow for the Washington Shore fish ladder.  The units are placed on standby from approximately 0000-0300 nightly if drawdown exceeds 1.0’.  Both fish units were out of service from 0800 until 1706 on 21 July due to mid season fishway dives at PH2.

1.5 Gatewell drawdowns are measured on Wednesdays.

2.  ACTIVITIES. 
2.1 AWS Closures.  The following AWS valves were closed this reporting week.

2.1.1 FV1-1, FV3-7, FV3-9, FV4-3, and FV4-4 from 1000 until 1335 on 22 July for mid season dive.

2.1.2 FV5-3, FV5-4, and FV5-9 from 0730 until 0930 on 22 July for mid season dive.

2.1.3 FV6-9 from 0800 until 1630 on 21 July for mid season dive.

2.2 STS/VBS Inspections.  No STS/VBS inspections were conducted this reporting week.

2.3 Dewatering and Fish Salvage.  No dewatering or fish salvage activities this reporting week.

3.  RESEARCH.

3.1 CRITFC.  Sampling has concluded until water levels drop below 70º F.

3.2 NMFS Lamprey.  Trapping, tagging and tracking continue.

3.3 ODFW Steelhead.  Trapping continues.  

3.4 PSMFC at DSM1.  Continuous PIT tag interrogation continues.  

3.5 PSMFC at the SMF.  GBT exams continue.

3.6 University of Idaho.  Adult tracking tagging and tracking continue.  Reported one dead fish. See misc. 5.4
3.7 U of I Transportation Study.  Evaluation of Snake River fall chinook will continue until 15 August.

3.8 University of Washington.  Northern pikeminnow tagging and tracking concluded. 

3.9 WDFW Fish Counting.  16-hr visual counts at Washington Shore and Bradford Island.

3.10 WES/Battelle/BAE systems Juvenile hydroacoustic monitoring.  Gap loss monitoring concluded.

4.  FISHWAYS 
4.1 Project biologists inspected 20 through 26 July this reporting week.

4.2 Fish Passage Plan Observations.

	DATE
	LOCATion
	FPP Deviation
	CAUSE
	RESPONSE
	TR/WAR

	All week
	PH1 PCC
	Velocity <1.5fps
	Meter in slowest part of channel
	No response at this time.
	

	All week
	PH2 PCC
	No velocity measure.
	Meter is missing
	New meter installed.  See misc. 5.1
	

	7/20-22
	PH1 ITS
	Chain gates 7A and 10C are closed.
	Headgate delivery.
	4B & 6B opened.  See misc. 5.7
	

	7/20
	Weir 37
	Reading <1.3’
	Unknown
	Operators to adjust.
	

	7/21
	WS fishway
	All gauges reading <1.0’. AWS valves closed.
	Mid season dives
	
	

	7/22
	Main dam and PH1 fishways.
	All gauges reading <1.0’. AWS valves closed
	Mid season dives
	
	

	7/23-26
	PH1 PCC
	WG2 open < 8’
	Breaker tripped
	See misc. 5.11
	


4.3 Fish Ladder Temperatures.  Measurements are taken three weirs down from the count station at Bradford Island and at the junction of the UMT and the main ladder at Washington shore.  Temperatures are in Kelvin.  http://students.washington.edu/kyle/temptable.jpg   (295.37K = 22.22C= 72F)

	Date
	Wash. Shore
	Bradford Is.
	AFF exit

	7/20
	294.2
	294.2
	294.3

	7/21
	294.7
	294.7
	294.7

	7/22
	294.8
	294.9
	294.9

	7/23
	294.6
	294.7
	294.5

	7/24
	294.4
	294.3
	294.4

	7/25
	294.3
	294.3
	294.3

	7/26
	294.2
	294.2
	294.2


4.4 Sensor Calibration.  Calibration was checked on 23 July this reporting week.

4.5  Zebra Mussel Monitoring.  No signs of colonization were observed this reporting week. 

5. MISCELLANEOUS

5.1 A new velocity meter was installed for PH2 PCC during the mid season dive on 21 July.  We are now waiting for the electricians to finish installation.

5.2 As of 28 June, the Project will have a daytime spill of 75 kcfs.

5.3 In a 10 July teletype the Project was given a hard constraint maximum tailwater of 16’ until 01 September.  This is for the B2CC work at PH2.  In a 15 July teletype, the maximum tailwater constraint overrules spill to the 120% gas cap if spill raises tailwater over 16’.

5.4 UI reported that an unclipped chinook mortality occurred while in transit to the Skamania release site on 21 July.  Please contact UI for more information.

5.5 Mid season dives were conducted on 21 July for PH2 and on 22 July for Cascades Island, B-Branch and PH1.  All diffuser grates and associated hardware were in good condition.  We did find a lot of dead lamprey below the diffuser grates at Cascades Island but not in any of the other fishways.  This will be discussed in the FPOM lamprey task group conference call on 28 July.  Video is available for anyone wishing to view it.

5.6 DSM1 went to manual operation at 2230 on 21 July due to a failure of the PLC to control properly.  The DSM returned to auto on 22 July.

5.7 Due to headgate delivery, chain gates 7A and 10C were closed (4B and 6B opened) on 10 July.  All chaingates returned to normal elevations on 22 July.

5.8 Remaining TIES were removed on 22 July.

5.9 STSs were installed in Unit 6 on 22 July to clear deck space for crane rail work starting 01 August.

5.10 Spill bay 11 was in manual and dogged at a one foot from 1700 on 23 July until 0800 on 24 July due to a broken power cord/plug.  Adjacent bays were used to compensate spill.

5.11 WG2 stopped controlling properly on 23 July.  The gate wouldn’t open to greater than about a four foot opening.  With WG2 raised so high, FV1-1 closed down.  Differential was maintained in the collection channel but velocities were well below 1.5fps.  Operators reset WG2 on 27 July.  The PH1 collection channel is back in criteria.

JAMES R. MAHAR, P.E.

Operations Manager
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